Author's response to reviews

Title: Dental Caries Status of Dai Preschool Children in Yunnan, China

Authors:

S N Zhang (s.nan1984@hotmail.com)
Edward CM Lo (hrdplcm@hkucc.hku.hk)
J Liu (liujuan1966@yahoo.com)
C H Chu (chchu@hku.hk)

Version: 3 Date: 23 July 2013

Author's response to reviews: see over
The Editor,
BMC Oral Health,
BioMed Central Ltd,
Floor 6, 236 Gray's Inn Road,
London, WC1X 8HL,
United Kingdom.

18 July 2013

Dear Editor,

Re: Submission of manuscript

We would like to submit our manuscript entitled **Dental Caries Status of Dai Preschool Children in Yunnan, China** for your consideration of publication in your journal. Dental caries remain a major public health concern in many countries, in particular the minority and disadvantaged community. Dai is a minority ethnic group but have more than 1.25 million people in China. There was little dental survey for Dai children in China, and this study can provide update information which can be achieved in WHO ORAL HEALTH country profiles. We sincerely hope you will review our manuscript.

Thank you for your attention.

Yours sincerely,

Dr C H Chu

Corresponding author